Author’s response to reviews

Title: Mucopolysaccharidosis Type VI: Case Report with First Neonatal Presentation with Ascites Fetalis and Rapidly Progressive Cardiac Manifestation

Authors:

Rachel Honjo (rachel.honjo@hc.fm.usp.br)
Evelyn Nuñez Vaca (cristinanunezvaca@gmail.com)
Gabriela Leal (gnleal@gmail.com)
Deipara Abellan (deipara.abellan@hc.fm.usp.br)
Nana Ikari (nana.miura@hc.fm.usp.br)
Marcelo Jatene (marcelo.jatene@hc.fm.usp.br)
Ana Martins (ni.martins10@gmail.com)
Chong Kim (chong.kim@hc.fm.usp.br)

Version: 4 Date: 08 Feb 2020

Author’s response to reviews:

São Paulo, Brazil, Feb 8th, 2020.

Dear Editorial Board of BMC Medical Genetics,

Please find below the answers to the last revision:

1. Consent to participate

Please remove the following statement in the ‘Ethics approval and consent to participate’ section of your manuscript: “The patient’s legal guardians (parents) have provided written consent to publish this case report, including medical data and images.”.

Answer: The statement was removed.

2. Consent for publication
Please change the following statement in the ‘Consent for publication’ section of your manuscript: “The patient’s legal guardians (parents) have signed informed consent to participate in this study and consented to publish this case report, including medical data and images.” to “The patient’s legal guardians (parents) have provided written consent to publish this case report, including medical data and images”.

Answer: The statement was replaced.

3. Availability of data and materials

Please include further detail in the Availability of data and materials section of your manuscript in regard to who should be contacted and their contact details or where to apply for permissions to obtain access to the raw data generated and/or analysed in your study.

Answer: The following sentence was added: For further details regarding this case report, please contact Prof Chong Ae Kim, MD, PhD (chong.kim@hc.fm.usp.br).

4. Figures 2, 3 and 4

As per our submission guidelines, we also require that all figures be individually cited in the main text in sequence in a manuscript. Figures 2, 3 and 4 are currently not cited in the main text in your manuscript. Please ensure that they are appropriately cited within the main body of text in your manuscript. Any items which are not cited may be deleted by our production department upon publication.

Answer: The Figures were cited individually in the modified manuscript, and the order was updated (Fig 2 and 4 had their numbers interchanged to correspond to the text in the manuscript). Figures were uploaded in the system with the new numbers.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Answer: The revision files are now updated in the submission system.

Kind regards,

Rachel Honjo, on behalf of all the authors